



University of Groningen

Effect of Canagliflozin on Renal and Cardiovascular Outcomes across Different Levels of Albuminuria

Neuen, Brendon L.; Ohkuma, Toshiaki; Neal, Bruce; Matthews, David R.; de Zeeuw, Dick; Mahaffey, Kenneth W.; Fulcher, Greg; Li, Qiang; Jardine, Meg; Oh, Richard

Published in: Journal of the American Society of Nephrology

DOI: 10.1681/ASN.2019010064

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2019

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Neuen, B. L., Ohkuma, T., Neal, B., Matthews, D. R., de Zeeuw, D., Mahaffey, K. W., Fulcher, G., Li, Q., Jardine, M., Oh, R., Heerspink, H. L., & Perkovic, V. (2019). Effect of Canagliflozin on Renal and Cardiovascular Outcomes across Different Levels of Albuminuria: Data from the CANVAS Program. *Journal of the American Society of Nephrology, 30*(11), 2229-2242. https://doi.org/10.1681/ASN.2019010064

Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

Take-down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.

Effect of Canagliflozin on Renal and Cardiovascular Outcomes across Different Levels of Albuminuria: Data from the CANVAS Program

Brendon L. Neuen,¹ Toshiaki Ohkuma,¹ Bruce Neal,¹ David R. Matthews,² Dick de Zeeuw,³ Kenneth W. Mahaffey,⁴ Greg Fulcher,⁵ Qiang Li,¹ Meg Jardine,¹ Richard Oh,⁶ Hiddo L. Heerspink,^{1,3} and Vlado Perkovic¹

¹The George Institute for Global Health, University of New South Wales, Sydney, Australia; ²Oxford Centre for Diabetes, Endocrinology and Metabolism, Harris Manchester College, University of Oxford, Oxford, UK; ³University Medical Center Groningen, University of Groningen, Groningen, The Netherlands; ⁴Department of Medicine, Stanford Center for Clinical Research, Stanford University School of Medicine, Stanford, California; ⁵Royal North Shore Hospital, Sydney, Australia; and ⁶Janssen Research & Development, LLC, Raritan, New Jersey

ABSTRACT

Background If SGLT2 inhibitors protect the kidneys by reducing albuminuria as hypothesized, people with type 2 diabetes mellitus (T2DM) with higher albuminuria should benefit more.

Methods We conducted a *post-hoc* analysis of data from the CANagliflozin cardioVascular Assessment Study (CANVAS) Program, which randomized 10,142 participants with T2DM and high cardiovascular risk to canagliflozin or placebo. We assessed effects of canagliflozin on renal, cardiovascular, and safety outcomes by baseline albuminuria. The trial included 2266 participants (22.3%) with moderately increased albuminuria (urinary albumin/creatinine ratio [UACR] 30–300 mg/g) and 760 (7.5%) with severely increased albuminuria (UACR > 300 mg/g) at baseline.

Results Canagliflozin lowered albuminuria with greater proportional reductions in those with moderately and severely increased albuminuria (*P* heterogeneity<0.001). After week 13, canagliflozin slowed the annual loss of kidney function across albuminuria subgroups, with greater absolute reductions in participants with severely increased albuminuria (placebo-subtracted difference 3.01 ml/min per 1.73 m² per year; *P* heterogeneity<0.001). Heterogeneity for the renal composite outcome of 40% reduction in eGFR, ESKD, or renal-related death was driven by lesser effects in participants with moderately increased albuminuria (*P* heterogeneity=0.03), but no effect modification was observed when albuminuria was fitted as a continuous variable (*P* heterogeneity=0.94). Cardiovascular and safety outcomes were mostly consistent across albuminuria levels including increased risks for amputation across albuminuria subgroups (*P* heterogeneity=0.66). Greater absolute risk reductions in the renal composite outcome were observed in participants with severely increased albuminuria (*P* heterogeneity=0.004).

Conclusions The proportional effects of canagliflozin on renal and cardiovascular outcomes are mostly consistent across patients with different levels of albuminuria, but absolute benefits are greatest among those with severely increased albuminuria.

JASN 30: 2229-2242, 2019. doi: https://doi.org/10.1681/ASN.2019010064

Received January 21, 2019. Accepted July 16, 2019.

Global Health, University of New South Wales Sydney, Level 5, 1 King Street, Newtown, Sydney, NSW 2042, Australia. Email: vperkovic@georgeinstitute.org.au

Published online ahead of print. Publication date available at www.jasn.org.

Correspondence: Dr. Vlado Perkovic, The George Institute for

Copyright © 2019 by the American Society of Nephrology

CKD is one of the leading causes of morbidity and mortality in type 2 diabetes mellitus (T2DM), developing in approximately 40% of affected individuals.^{1,2} Albuminuria is one of the earliest clinically detectable manifestations of kidney damage, and is an independent risk factor for cardiovascular events, kidney failure, and death.^{3,4} Efforts to prevent these outcomes have targeted not only BP and glucose control, but also the lowering of albuminuria with renin-angiotensin system (RAS) blockade, which has been associated with subsequent renoprotection.^{5,6}

Canagliflozin is a sodium glucose cotransporter 2 (SGLT2) inhibitor which promotes glycosuria and natriuresis, resulting in reductions in glycated hemoglobin (HbA1c), BP, and body weight.⁷ Canagliflozin and other SGLT2 inhibitors also ameliorate albuminuria, resulting in an approximate one-third reduction in albuminuria in people with moderately or severely increased albuminuria.⁸ These multiple metabolic benefits have translated into a reduction in cardiovascular events in large cardiovascular outcome trials.^{9–13}

Most recently, the Canagliflozin and Renal Events in Diabetes with Established Nephropathy Clinical Evaluation (CREDENCE) trial demonstrated that canagliflozin reduces the risk of kidney failure by approximately 30% in people with established diabetic kidney disease.¹⁴ Almost all participants in the CREDENCE trial had severely increased albuminuria at baseline with a median urinary albumin/creatinine ratio (UACR) of 927 mg/g.¹⁴ There is therefore uncertainty as to whether the renal and cardiovascular benefits demonstrated in the CREDENCE trial are generalizable across a wider range of albuminuria, especially to people with T2DM and lesser degrees of albuminuria.

We hypothesized that by reducing albuminuria, SGLT2 inhibitors might be particularly beneficial for renal and perhaps also cardiovascular outcomes in people with T2DM and higher levels of albuminuria. We therefore undertook a range of *post-hoc* analyses of the CANagliflozin cardioVascular Assessment Study (CANVAS) Program to determine the effect of canagliflozin on renal, cardiovascular, and safety outcomes in people with T2DM according to baseline levels of albuminuria.

METHODS

Study Design and Participants

The detailed methods and statistical analysis plan for the CANVAS Program have been published previously.⁹ Briefly, the CANVAS Program comprised two multicenter, doubleblind, placebo-controlled randomized trials (CANVAS [NCT01032639] and CANVAS-R [NCT01989754]) with identical key inclusion criteria that were designed to assess the cardiovascular safety and efficacy of the SGLT2 inhibitor, canagliflozin, along with effects on renal and safety outcomes in people with T2DM at high cardiovascular risk. The trials were conducted in 667 centers across 30 countries. Local

Significance Statement

Albuminuria commonly occurs in people with type 2 diabetes and is an independent risk factor for progression of kidney disease and cardiovascular events. SGLT2 inhibitors are thought to protect the kidneys by lowering albuminuria. If this is true, it suggests people with type 2 diabetes with higher levels of albuminuria would reap greater renoprotective benefits. The authors conducted a *post-hoc* analysis of data from the CANagliflozin cardioVascular Assessment Study (CANVAS) Program to assess renal, cardiovascular, and safety outcomes with canagliflozin by baseline albuminuria subgroups (urinary albumin/creatinine ratio <30, 30–300, and >300 mg/g). The data suggest that the relative effects of canagliflozin on renal and cardiovascular outcomes are mostly consistent across different levels of baseline albuminuria, but participants with severely increased albuminuria saw the largest absolute benefits.

institutional ethics committees approved the trial protocols at each site, and all participants provided written informed consent.

The trials included participants with T2DM and HbA1c levels \geq 7.0% and \leq 10.5% who were either 30 years or older with established atherosclerotic vascular disease, or 50 years or older with two or more cardiovascular risk factors. These risk factors included: duration of diabetes of at least 10 years, systolic BP higher than 140 mm Hg while receiving one or more antihypertensive agents, UACR of at least 30 mg/g, current smoking, or HDL cholesterol level of <1 mmol/L. Participants with a baseline eGFR <30 ml/min per 1.73 m² were excluded.

Randomization and Masking

All potentially eligible participants underwent a 2-week, single-blind, placebo run-in period before randomization. Randomization procedures differed between the trials. Participants in CANVAS were randomly assigned in a 1:1:1 ratio to receive canagliflozin 100 mg daily, canagliflozin 300 mg daily, or placebo; whereas participants in CANVAS-R were randomly assigned in a 1:1 ratio to receive canagliflozin 100 mg daily or matching placebo, with an optional increase to 300 mg or matching placebo daily starting from week 13. Randomization was performed centrally through a web-based response system with the use of a computer-generated randomization schedule with randomly permuted blocks that were prepared by the trial sponsor. All participants and trial staff were blinded to individual treatment allocations until the end of the trial.

Follow-Up Procedures

Face-to-face follow-up was scheduled at least three times in the first year and at intervals of 6 months thereafter. Serum creatinine was measured at least three times in the first year including at baseline, and then every 26 weeks. UACR was measured in first-morning void urine specimens at baseline and at week 12 and then annually in CANVAS, and every 26 weeks in CANVAS-R. Adverse event assessment was performed at each visit. Other glycemic and cardiovascular risk factor management, including RAS blockade, was guided by best practice in accordance with local guidelines.

Outcomes

Definitions for all clinical outcomes in the CANVAS Program have been previously published.9 The primary outcome was a composite of cardiovascular death, nonfatal myocardial infarction, or nonfatal stroke. Other outcomes included cardiovascular death, fatal or nonfatal myocardial infarction, fatal or nonfatal stroke, hospitalized or fatal heart failure, and allcause mortality. The renal outcomes of interest in this analysis were: (1) mean annualized difference in eGFR between canagliflozin and placebo; and (2) a sustained and independently adjudicated composite of 40% reduction in eGFR, ESKD, or renal death. The end point of 40% reduction in eGFR was sent for adjudication if sustained for two consecutive measures \geq 30 days apart or occurring on the last available measure. eGFR was calculated using the Modification of Diet in Renal Disease equation using centrally measured serum creatinine collected at study visits. Central end point adjudication committees blinded to treatment allocation assessed cardiovascular, renal, and key safety outcomes.

Adverse events, both serious and nonserious, were collected in the CANVAS trial until early 2014, as mandated by the regulatory agencies as a requirement for initial approval for the use of canagliflozin. After this time, only serious adverse events, adverse events leading to study drug discontinuation, and selected adverse events of interest were collected in the CANVAS trial. This streamlined adverse event collection approach was used for the entirety of CANVAS-R. We therefore reported all adverse events for the CANVAS trial separately, along with all serious adverse events across the CANVAS Program.

Statistical Analyses

Baseline characteristics for participants with normal, moderately increased, and severely increased albuminuria at baseline (defined as UACR < 30, 30–300, and > 300 mg/g, respectively) were compared using chi-squared and ANOVA tests for categorical and continuous variables.

The effects of canagliflozin on intermediate outcomes and eGFR over time were calculated from baseline to week 312. The mean change in HbA1c, BP, and body weight over time and the difference between canagliflozin and placebo were analyzed using mixed effect models for repeated measurements that included all of the postbaseline data up to week 312 and the covariates for study, visit, treatment, baseline measures, and baseline-by-visit interactions. Because of the highly skewed distribution of UACR data, UACR data were log-transformed and the geometric mean of postbaseline UACR was estimated using a similar mixed effect model. Changes in albuminuria were calculated as the ratio of the geometric mean of postrandomization UACR measures with canagliflozin compared with placebo. We also included treatment by visit as a covariate when assessing the effect of canagliflozin on eGFR over time.

Because of the recognized nonlinear association between eGFR and time resulting from the acute hemodynamic effect of canagliflozin, the differences in eGFR slope between canagliflozin and placebo were assessed by a piecewise linear mixed effect model using an intention-to-treat approach over the total study duration and separately in two time periods: baseline to week 13 (acute slope), and week 13 to last available measures during the trial period (chronic slope). A time-spline variable measuring the follow-up time from week 13 was introduced in the model to accommodate the nonlinear trends of the eGFR time trajectory. eGFR data collected at the scheduled visits were regressed by the fixed effects with terms for treatment and study, and with linear covariates of time, time spline, and interactions of treatment by time and treatment by the spline variable. Intercept, time, and time spline were included as random effects to allow variation between participants. Time covariates included in the model were calculated in years to estimate annualized changes in eGFR.

The effects of canagliflozin on cardiovascular, renal, and safety outcomes were analyzed overall and in participants with normal, moderately increased, and severely increased albuminuria. Hazard ratios (HRs) and 95% CIs for the cardiovascular outcomes were estimated with Cox regression models for all canagliflozin groups combined versus placebo, with stratification according to trial and history of cardiovascular disease. The same method was used for the composite renal outcome, with adjustment for baseline eGFR (<60 or \geq 60 ml/min per 1.73 m²) and stratification by trial. Analyses for cardiovascular and renal outcomes were conducted on the full integrated data set using an intention-to-treat approach, which included all events that occurred at any time from randomization to the last follow-up date; participants were censored at the time of cardiovascular or noncardiovascular death or last trial contact date for those not experiencing an event or lost to follow-up. For the renal composite outcome, participants were censored when renal- or nonrenal-related deaths occurred, but not censored if any nonfatal types of events occurred. Annualized incidence rates were calculated per 1000 patient-years of follow-up. Sensitivity analyses adjusting for competing risk of death were performed for these outcomes using the Fine and Gray method.¹⁵

For safety outcomes, an on-treatment analysis was performed (using only events that occurred among participants who had a safety outcome while they were receiving canagliflozin or placebo, or within 30 days after discontinuation of drug or placebo). For amputation and fracture outcomes, analyses included participants who received at least one dose of canagliflozin or placebo and had an event at any time during follow-up.

The methods for determining heterogeneity of treatment effect involved adding UACR as a covariate and a term for UACR-by-treatment interaction to the relevant model to test for heterogeneity across albuminuria subgroups. Terms for UACR-by-time interaction were also included in the piecewise linear mixed model. The *P* heterogeneity values across subgroups were obtained using the likelihood ratio test. We explored heterogeneity further in a range of sensitivity analyses by testing for trend across ordered UACR subgroups, performing interaction tests across deciles of baseline UACR, and assessing for effect modification using log-transformed UACR fitted as a continuous variable. Tests for trend in treatment effect across subgroups were performed using a similar approach as described above with UACR subgroups treated as ordered categories.

Absolute effects on select outcomes of interest per 1000 patients over 5 years and corresponding 95% CIs were estimated as the differences in the incidence rates between randomized treatment groups, using Poisson regression analysis with an assumption of constant annual event probabilities. We performed three pairwise comparisons across normal, moderately increased, and severely increased albuminuria subgroups and reported the lowest (*i.e.*, most conservative) P value to test for heterogeneity.

Analyses were performed with SAS software version 9.2, SAS Enterprise Guide version 7.11, and STATA software version 15.1.

Data from the CANVAS Program will be made available in the public domain *via* the Yale University Open Data Access Project (http://yoda.yale.edu/) once the product and relevant indications studied have been approved by regulators in Europe and the United States and the study has been completed for 18 months.

RESULTS

The CANVAS Program randomized 10,142 participants, 10,033 (98.9%) of whom had UACR measured at baseline. A total of 9734 participants (96%) completed the study, with a mean follow-up of 188.2 weeks, although mean length of follow-up was longer in CANVAS (296 weeks) than in CANVAS-R (108 weeks). At baseline, there were 7007 participants with normal albuminuria (69.1%), 2266 (22.3%) with moderately increased albuminuria. Baseline use of RAS blockade was high overall and higher across subgroups with increasing albuminuria, ranging from 79.2% to 82.6%.

Baseline characteristics of canagliflozin- and placebotreated participants were mostly similar within each albuminuria subgroup (Table 1). Characteristics of participants by baseline albuminuria subgroups are also presented in Supplemental Table 1. Across progressively higher levels of albuminuria, participants were more likely to be male and have a longer duration of diabetes, higher systolic BP, and higher HbA1c (all P<0.001; Supplemental Table 1). The proportion of participants with established microvascular complications and a history of peripheral vascular disease and amputations increased with baseline albuminuria (all P<0.001; Supplemental Table 1). Mean eGFR was progressively lower across albuminuria subgroups, and participants were more likely to be treated with diuretics and insulin, and less likely to receive metformin and sulfonylureas (all P<0.001; Supplemental Table 1).

Intermediate Outcomes

The placebo-subtracted differences in HbA1c, systolic BP, body weight, and albuminuria varied in participants with normal, moderately increased, and severely increased albuminuria, likely due to differences in baseline eGFR across the subgroups (Figure 1). Proportional reductions in albuminuria increased with higher levels of albuminuria (P heterogeneity<0.001). Reductions in HbA1c were attenuated across progressively higher levels of albuminuria, with a similar pattern observed for body weight (P heterogeneity=0.002 and 0.08, respectively). In contrast, the effect on systolic BP was consistent across albuminuria subgroups (P heterogeneity=0.26). The effects on intermediate outcomes, displayed separately in canagliflozin- and placebo-treated participants, are summarized in Supplemental Figure 1.

Renal Outcomes

The effect of canagliflozin on eGFR slope varied during followup (Figure 2). Within 13 weeks, participants randomized to canagliflozin experienced a fall in eGFR, which was similar across subgroups with normal, moderately increased, and severely increased albuminuria (placebo-subtracted differences of -2.31, -2.50, and -2.73 ml/min per 1.73 m², respectively; P heterogeneity=0.66). From week 13 until the end of followup (*i.e.*, the chronic slope), canagliflozin attenuated the loss of kidney function across all levels of baseline albuminuria (Figure 2), with clear differences in the size of the treatment effect across subgroups (*P* heterogeneity<0.001). The annual rate of eGFR decline in placebo arms was progressively greater among higher categories of albuminuria (Figure 2). The corresponding annual mean difference in eGFR slope between canagliflozin and placebo was greatest in participants with severely increased albuminuria (placebo-subtracted difference of 3.01 ml/min per 1.73 m² per year) and lesser in those with normal and moderately increased albuminuria (placebo-subtracted differences of 1.06 and 0.99 ml/min per 1.73 m^2 per year, respectively; Figure 2). Total eGFR slope for the overall population and by albuminuria subgroups is summarized in Supplemental Table 2.

Heterogeneity was observed for the renal composite outcome across normal albuminuria (HR, 0.50; 95% CI, 0.33 to 0.77), moderately increased albuminuria (HR, 0.98; 95% CI, 0.60 to 1.60), and severely increased albuminuria subgroups (HR, 0.48; 95% CI, 0.31 to 0.74; P heterogeneity=0.03; Figure 3). The effect on the renal composite outcome adjusted for competing risk of death was similar, and is presented in Supplemental Table 3. However, there was no evidence of a trend in treatment effect across ordered albuminuria subgroups or when interaction tests were performed with log-transformed UACR fitted as a continuous variable (P trend=0.80 and *P* heterogeneity=0.94, respectively; Supplemental Table 4). The effect on the renal composite outcome was also consistent when participants were categorized into deciles based on UACR (*P* heterogeneity=0.33 and *P* trend=0.32; Supplemental Figure 2).

Characteristics

	Canagliflozin (<i>n</i> =4012)	Placebo (<i>n</i> =2995)	Canagliflozin (<i>n</i> =1322)	Placebo (<i>n</i> =944)	Canagliflozin (<i>n</i> =406)	Placebo (<i>n</i> =354)
Age, years, mean (SD)	63.0 (8.2)	63.2 (8.1)	63.8 (8.3)	64.2 (8.3)	63.4 (8.3)	64.0 (8.2)
Sex, no. (%)						
Male	2497 (62.2)	1835 (61.3)	936 (70.8)	638 (67.6)	284 (70.0)	248 (70.1)
Female	1515 (37.8)	1160 (38.7)	386 (29.2)	306 (32.4)	122 (30.0)	106 (29.9)
Race, no. (%)						
White	3129 (80.0)	2393 (79.9)	1030 (77.9)	722 (76.5)	301 (74.1)	272 (76.8)
Asian	527 (13.1)	323 (10.8)	187 (14.1)	137 (14.5)	62 (15.3)	47 (13.3)
Black	123 (3.1)	108 (3.6)	37 (2.8)	33 (3.5)	12 (3.0)	17 (4.8)
Other ^a	233 (5.8)	171 (5.7)	68 (5.1)	52 (5.5)	31 (7.6)	18 (5.1)
Current smoker, no. (%)	691 (17.2)	547 (18.3)	235 (17.8)	166 (17.6)	80 (19.7)	57 (16,1)
History of hypertension, no. (%)	3528 (87.9)	2682 (89.5)	1229 (93.0)	868 (91,9)	377 (92.9)	337 (95.2)
History of heart failure, no. (%)	526 (13.1)	439 (14.7)	200 (15.1)	149 (15.8)	63 (15.5)	59 (16.7)
Duration of diabetes, years, mean (SD)	12.9 (7.6)	13.2 (7.7)	14.4 (7.8)	14.3 (7.9)	15.4 (7.6)	16.0 (8.0)
Drug therapy, no. (%)			(,		,	
Insulin	1862 (46.4)	1405 (46.9)	738 (55.8)	526 (55.7)	266 (65.5)	248 (70.1)
Sulfonvlurea	1779 (44-3)	1310 (43 7)	586 (44.3)	388 (41 1)	145 (35 7)	118 (33 3)
Metformin	3120 (77.8)	2383 (79.6)	1022 (77.3)	705 (74 7)	260 (64 0)	243 (68 6)
GLP-1 receptor agonist	152 (3.8)	122 (4 1)	49 (3 7)	46 (4 9)	18 (4 4)	15 (4 2)
DPP-4 inhibitor	487 (12 1)	373 (12 5)	144 (10.9)	134 (14 2)	58 (14.3)	52 (14 7)
Statin	2978 (74.2)	2236 (74 7)	1007 (76.2)	727 (77 0)	305 (75.1)	272 (76.8)
Antithrombotic	2914 (72.6)	2226 (74.7)	977 (73.9)	705 (74 7)	306 (75.4)	272 (70.0)
RAAS inhibitor	3171 (72.0)	2378 (79.4)	1095 (82.8)	751 (79.6)	332 (81.8)	296 (83.6)
B-Blocker	2075 (51.7)	1654 (55.2)	726 (54.9)	513 (54 3)	214 (52 7)	188 (53 1)
Diuretic	1667 (41.6)	1271 (42 4)	644 (48 7)	452 (47.9)	200 (49 3)	202 (57 1)
Microvascular disease history no. (%)	1007 (11.0)	1271 (12.1)	011(10.7)	452 (47.77)	200 (17.0)	202 (07.17)
Retinopathy	735 (18 3)	563 (18.8)	319 (24 1)	230 (24.4)	133 (32.8)	124 (35-1)
Nephropathy	434 (10.8)	359 (12.0)	370 (28.0)	244 (25.8)	178 (43.8)	169 (47 9)
Neuropathy	1136 (28.3)	892 (29.8)	477 (36 1)	282 (29.9)	154 (37.9)	127 (35.9)
Atherosclerotic vascular disease history no. (%) ^b	1130 (20.3)	072 (27.0)	477 (30.1)	202 (27.7)	134 (37.7)	127 (55.7)
Coronany	2239 (55.8)	1754 (58.6)	751 (56.8)	517 (5/1.8)	214 (52 7)	190 (53 7)
Cerebrovascular	748 (18.6)	571 (19 1)	271 (20.5)	177 (18.8)	83 (20 /)	89 (25.1)
Peripheral	762 (19.0)	601 (20 1)	291 (22.0)	212 (22 5)	114 (28.1)	113 (31.9)
Any	2846 (70.9)	2218 (7/L 1)	955 (72.2)	673 (71 3)	287 (70 7)	271 (76 6)
Cardiovascular disease history no (%) ^c	2592 (64, 6)	2008 (67 0)	863 (65 3)	610 (64 6)	266 (65 5)	253 (71 5)
History of amputation no. (%)	62 (1 5)	2000 (07:0) 40 (1 3)	45 (3.4)	28 (3 0)	28 (6 9)	32 (9 0)
Body mass index kg/m ² mean (SD)	31.9 (5.9)	31.8 (5.9)	32 1 (5 9)	32 5 (6 1)	32 1 (6 1)	31.8 (6.0)
Systolic BP mm Ha mean (SD)	134.6 (15.1)	135.2 (15.1)	139 / (15 6)	139.2 (16.2)	145 1 (18 5)	1/1 8 (16 8)
Diastolic BP, mm Hg, mean (SD)	77.3 (9.4)	77 5 (9 5)	77.8 (10.0)	78 / (10.1)	80.1 (10.0)	79.0 (9.8)
Glucated homoglabin % maan (SD)	8 2 (0 9)	8 2 (0 9)	8.4 (1.0)	8.4.(0.9)	8 5 (1 0)	8 4 (0 9)
Total cholostoral mmol/L moan (SD)	4.3 (1.1)	0.2 (0.7)	0.4 (1.0) 4 4 (1.1)	0.4 (0.7) 4 3 (1 2)	4.7 (1.3)	0.4 (0.7) 4 5 (1 A)
Trichucerides mmol/L mean (SD)	4.3 (1.1)	4.4 (1.1)	4.4 (1.1)	4.3 (1.2)	4.7 (1.3)	4.3 (1.4)
HDL shalastaral mmal/L maan (SD)	1.7 (1.2)	1.7 (1.4)	2.2 (1.4)	2.2 (1.7)	2.4 (1.6)	2.3 (1.7)
IDL cholostorol mmol/L moon (SD)	2.3 (0.9)	2 3 (0 0)	2 3 (0 9)	2.2 (0.0)	2.5 (1.1)	2 / (1 1)
	2.3 (0.7)	2.3 (0.7)	2.3 (0.7)	2.2 (0.7)	2.3 (1.1)	2.4 (1.1)
CEP m/min nor 1.72 m ² moon (SD)	2.0 (0.7) 79.4 (10.5)	Z.U (U.7)	2.1 (U.7)	2.0 (0.7) 72.0 (21.0)	2.3 (1.1)	
UACP mg/g mgdian (intersectile reces)	/ 0.4 (17.3) 0 / (E 7 13 /)	/0.U (17.7) 0.2 /E 7 12.2)	/ 4.0 (20.7) 47 1 (40 4 107 0)	/ 3.7 (21.7) 40 / (// 4 120 E)	03.7 (22.2) 401 0 (422 0 1055 4)	00.7 (ZZ.3)
oAGN, mg/g, median (interquartile range)	0.4 (3.7-13.4)	0.2 (3./-13.2)	07.1 (42.0-127.2)	07.4 (44.0-120.3)	071.7 (433.2-1233.4)	103.2 (431.5-1394.1)

Table 1. Characteristics of canagliflozin- and placebo-treated participants by baseline albuminuria subgroups (UACR <30, 30-300, and >300 mg/g)

Normal Albuminuria

(<30 mg/g)

Moderately Increased Albuminuria

(30–300 mg/g)

Severely Increased Albuminuria

(>300 mg/g)

^aIncludes American Indian or Alaska Native, Native Hawaiian or other Pacific Islander, multiple, other, and unknown.

 $^{\rm b}{\rm Some}$ participants had >1 type of atherosclerotic disease.

^cAs defined in the protocol.



Figure 1. Placebo-subtracted differences in intermediate outcomes varied in participants with UACR <30, 30–300, and >300 mg/g at baseline.

Cardiovascular Outcomes

The relative effect of canagliflozin on the primary cardiovascular composite outcome in the overall trial population (HR, 0.86; 95% CI, 0.75 to 0.97) was consistent across albuminuria subgroups (*P* heterogeneity=0.46; Figure 3). The same was true for all other cardiovascular outcomes, as well as all-cause mortality (all *P* heterogeneity>0.20; Figure 3). Results were similar in sensitivity analyses adjusting for the competing risk of death (Supplemental Table 3). No heterogeneity was observed for any of these outcomes when interaction tests were undertaken testing for trends across ordinal subgroups, or using log-transformed UACR fitted as a continuous variable (Supplemental Table 4). Similarly, effects on the primary cardiovascular outcome, heart failure, and all-cause mortality were consistent across deciles of UACR (Supplemental Figure 2).

Absolute Effects

Absolute risk differences for canagliflozin versus placebo across different levels of albuminuria are summarized in Figure 4. The

absolute risk reductions for most cardiovascular outcomes were consistent across albuminuria subgroups. Heterogeneity was observed for the renal composite outcome and for allcause mortality; absolute risk reductions for these outcomes were larger in people with severely increased albuminuria (P heterogeneity=0.004 and 0.04, respectively). There was no statistically significant evidence of heterogeneity for the absolute effect on amputations (P heterogeneity=0.11), although the risk appeared numerically greater in the severely increased albuminuria subgroup, which had a higher proportion of participants with a prior history of amputations.

Safety Outcomes

The risks of most adverse outcomes with canagliflozin, including renal safety outcomes, were consistent across albuminuria subgroups (Figure 5, Supplemental Figure 3). The increased risk of amputation with canagliflozin observed in the overall trial population was consistent across albuminuria



Figure 2. Canagliflozin slowed the loss of kidney function across all UACR subgroups. Effect of canagliflozin as indicated by (A) adjusted mean eGFR over time and (B) eGFR slope from week 6/13 until the end of follow-up, with the greatest effect in participants with UACR >300 mg/g at baseline. *Data are reported for week 6 in CANVAS and week 13 in CANVAS-R.

Patients with an event per

1000 patient-years		t-years			
	Canagliflozin	Placebo		HR (95% CI)	P heterogeneity
MACE					0.46
All	26.9	31.5	⊢ ♠⊣	0.86 (0.75 to 0.97)	
<30	22.1	26.5	HO-K	0.83 (0.71 to 0.98)	
30 to 300	35.2	35.4	H-Q-4	0.98 (0.76 to 1.25)	
>300	53.6	72.0	⊢ o i	0.75 (0.53 to 1.06)	l.
CV death					0.62
All	11.6	12.8	⊢ ♦ 1	0.87 (0.72 to 1.06)	
<30	8.6	9.1	H-O+-I	0.91 (0.70 to 1.19)	
30 to 300	16.0	15.8	⊢_dI	0.98 (0.69 to 1.41)	
>300	31.3	42.6		0.70 (0.45 to 1.07)	l.
Fatal/nonfa	tal MI				0.38
All	11.2	12.6	⊢ ♣∔1	0.89 (0.73 to 1.09)	
<30	9.9	11.4	⊢-œ÷-i	0.89 (0.69 to 1.13)	
30 to 300	14.7	13.7	⊢́pI	1.05 (0.71 to 1.56)	
>300	14.0	23.2		0.62 (0.33 to 1.16)	
Fatal/nonfa	tal stroke				0.50
All	7.9	9.6	⊢ ♦ 1	0.87 (0.69 to 1.09)	
<30	6.3	8.4		0.76 (0.56 to 1.02)	
30 to 300	9.5	10.8		0.92 (0.59 to 1.46)	
>300	20.6	19.0		1.23 (0.67 to 2.26)	1
Hospitalize	d or fatal HF				0.96
All	6.4	9.7	⊢	0.70 (0.55 to 0.89)	
<30	4.3	6.6		0.67 (0.47 to 0.94)	
30 to 300	9.3	12.9		0.74 (0.48 to 1.15)	
>300	22.1	33.3		0.73 (0.43 to 1.22)	
Renal comp	osite*				0.03
All	5.5	9.0	⊢-♦1	0.60 (0.47 to 0.77)	
<30	2.6	4.9		0.50 (0.33 to 0.77)	
30 to 300	9.0	8.4	н <u>ф</u>	0.98 (0.60 to 1.60)	
>300	29.4	56.6		0.48 (0.31 to 0.74)	
All-cause m	ortality				0.21
All	17.3	19.5	⊢ ♠-Ì	0.87 (0.74 to 1.01)	
<30	13.7	14.8	H-O+I	0.89 (0.72 to 1.10)	
30 to 300	23.5	22.9	⊢¢1	1.00 (0.74 to 1.34)	
>300	37.3	57.5		0.63 (0.43 to 0.92)	
			· · · · · · · · · · · · · · · · · · ·		
			0.25 0.5 1.0 2.0 4.0)	
			Favors canagliflozin Favors placebo		

Figure 3. There was heterogeneity in the effect of canagliflozin on the renal composite outcome in participants with UACR <30, 30–300, and >300 mg/g at baseline, while effects on cardiovascular outcomes and all-cause mortality were consistent across UACR subgroups. *A 40% reduction in eGFR, ESKD, or renal death. CV, cardiovascular; HF, heart failure; HR, hazard ratio; MACE, major adverse cardiovascular events; MI, myocardial infarction.

subgroups (*P* heterogeneity=0.66). There was some evidence of heterogeneity for urinary tract infections and fractures (*P* heterogeneity=0.04 and 0.07, respectively). Heterogeneity for the fracture outcome remained when fitting UACR as a continuous variable (*P* heterogeneity=0.03). There was no evidence of heterogeneity in the risk of urinary tract infections when analyzing UACR as a continuous variable (*P* heterogeneity=0.22).

DISCUSSION

In this analysis of the CANVAS Program, canagliflozin slowed the loss of kidney function, as measured by eGFR slope, at all levels of albuminuria. The absolute effect on eGFR slope was approximately three times as large in people with severely increased albuminuria, and this was due, at least in part, to the

Number of patients with an event		Patients with an event per 1000 patient-years		e	Excess number of active patients experiencing the yeart in 1000 patients over	
		Canagliflozin	Placebo		5 years (95% Cl)*	
MACE						0.12
All	1011	26.9	31.5	H+H	-23 (-41 to -4)	
<30	596	22.1	26.5	юн	-22 (-42 to -2)	
30 to 3	00 272	35.2	35.4	ю на на на на на на на на на на на на на	-1 (-44 to 42)	
>300	134	53.6	72.0		-92 (-200 to 15)	
Hospit	alized or fatal HF					0.23
All	276	6.4	9.7	I ♦I	-17 (-26 to -7)	
<30	133	4.3	6.6	D	-12 (-21 to -2)	
30 to 3	00 84	9.3	12.9	ноні	-18 (-43 to 6)	
>300	59	22.1	33.3		-56 (-127 to 16)	
Renal	composite [†]					0.004
All	249	5.5	9.0	I	-17 (-27 to -8)	
<30	89	2.6	4.9		-12 (-19 to -4)	
30 to 3	00 70	9.0	8.4	нфн	3 (-18 to 24)	
>300	88	29.4	56.6 H		-136 (-227 to -45)	
All-cau	use mortality					0.04
All	681	17.3	19.5	l ⊕ i	-11 (-25 to 3)	
<30	375	13.7	14.8	нсн	-5 (-20 to 10)	
30 to 3	00 193	23.5	22.9	нġн	3 (-31 to 37)	
>300	108	37.3	57.5		-101 (-191 to -11)	
Lower	extremity amputati	on				0.11
All	187	6.3	3.4	I	15 (8 to 22)	
<30	90	4.4	2.0		12 (5 to 19)	
30 to 3	00 61	8.9	5.7	Нон	16 (-3 to 35)	
>300	34	19.8	9.4	j—o	52 (3 to 102)	
			_250 ←	-150 -50 0 50	150	

Figure 4. Absolute benefits per 1000 participants over 5 years with canagliflozin versus placebo were consistent across UACR subgroups for cardiovascular outcomes, but absolute risk reductions for the renal composite outcome and all-cause mortality were greatest in patients with UACR >300mg/g. *Excess number is relative to the placebo group. If the number is negative, then fewer participants in the canagliflozin group experienced the event compared with the placebo group. [†]A 40% reduction in eGFR, ESKD, or renal death. HF, heart failure; MACE, major adverse cardiovascular events.

much more rapid decline in kidney function in this group. Kidney function loss was essentially completely abrogated in canagliflozin-treated participants with normal or moderately increased albuminuria. Although there was some evidence of heterogeneity for the renal composite outcome across normal, moderately increased, and severely increased albuminuria subgroups, the unusual pattern of effect and the absence of heterogeneity in a range of sensitivity analyses, including when interaction tests were performed with albuminuria fitted continuously, suggest that this may have been a chance finding. While the largest absolute renal benefits were observed in participants with severely increased albuminuria, the evidence of renoprotection, even among participants with normal albuminuria, suggests that mechanisms other than those associated with albuminuria reduction might also be important.

The effects of SGLT2 inhibition have now been studied across a spectrum of urinary albumin excretion. Our findings are broadly consistent with those from the EMPA-REG OUTCOME trial, in which approximately 60% of participants had normal albuminuria at baseline. The EMPA-REG OUTCOME trial demonstrated that the effects of empagliflozin on cardiovascular and renal outcomes were consistent across different levels of albuminuria, with greater

			_		
	Canagliflozin	Placebo		HR (95% CI) Ph	eterogeneity
All serious	adverse events				0.48
All	129.5	146.0		0.93 (0.87 to 1.00)	
<30	117.2	128.9	_	0.95 (0.87 to 1.03)	
30 to 300	147.8	176.5		0.88 (0.76 to 1.01)	
>300	212.4	256.7	ней	0.88 (0.70 to 1.09)	
Adverse ev	vents leading to c	discontinua	tion		0.13
All	35.7	32.9	•	1.13 (0.99 to 1.28)	
<30	32.5	29.0	ЮЛ	1.16 (0.99 to 1.37)	
30 to 300	39.7	32.7	Ŕон	1.26 (0.96 to 1.65)	
>300	60.9	77.3	нон	0.80 (0.56 to 1.15)	
Lower extr	emity amputatior	า			0.66
All	6.3	3.4	⊢◆-1	1.97 (1.41 to 2.75)	
<30	4.4	2.0	⊢⊡⊸∣	2.24 (1.35 to 3.70)	
30 to 300	8.9	5.7	ii-oi	1.53 (0.87 to 2.69)	
>300	19.8	9.4		2.36 (1.09 to 5.08)	
All fracture	s				0.07
All	15.4	11.9	I ♠I	1.26 (1.04 to 1.52)	
<30	15.7	10.4	ю	1.44 (1.14 to 1.83)	
30 to 300	14.2	15.5	нġн	0.93 (0.63 to 1.35)	
>300	17.4	16.9	ніськи страни	1.05 (0.55 to 2.02)	
Serious re	nal-related advers	se events			0.74
All	2.5	3.3	⊢ ♦ ∔I	0.76 (0.49 to 1.19)	
<30	1.8	2.6		0.68 (0.37 to 1.25)	
30 to 300	3.7	3.7	н-ф	0.95 (0.41 to 2.19)	
>300	7.7	9.1		0.85 (0.31 to 2.37)	
Serious ac	ute kidney injury				0.45
All	1.6	2.5	⊢ ♦ H	0.66 (0.39 to 1.11)	
<30	1.1	2.1		0.52 (0.25 to 1.06)	
30 to 300	2.7	2.5	ь р ц	1.09 (0.40 to 2.98)	
>300	4.8	6.5		0.69 (0.20 to 2.40)	
Serious hy	perkalemia				0.23
All	0.4	0.6	⊢	0.75 (0.27 to 2.11)	
<30	0.2	0.5		0.32 (0.06 to 1.81)	
30 to 300	1.2	0.4	н <u></u>	3.47 (0.40 to 30.22)	
>300	1.0	2.6 F	Ci	0.35 (0.03 to 3.89)	
			0.125 0.25 0.5 1.0 2.0 4.0 8.0 1	16.0 32.0	
			Favors canagliflozin Favors placebo	7	

Patients with an event per 1000 patient-vears

Figure 5. The risk of most safety outcomes collected across the CANVAS Program was consistent in participants with UACR <30, 30–300, and >300 mg/g at baseline. HR, hazard ratio.

effects on eGFR slope in participants with severely increased albuminuria.^{16–18} The CREDENCE trial demonstrated clear reductions in the risk of kidney failure and cardiovascular events in a population with severely increased albuminuria. Taken together, the data suggest that SGLT2 inhibitors are likely to confer cardiovascular and kidney benefits across a range of albuminuria, with greater absolute benefits in those with higher levels of urinary albumin excretion.

A number of potential mechanisms of renoprotection with SGLT2 inhibitors have been postulated.8 The acute decrease in eGFR with canagliflozin followed by long-term preservation of kidney function has been postulated to indicate decompression of the glomerulus and correction of glomerular hyperfiltration.^{8,19} Hyperfiltration is critical in the pathogenesis of diabetic kidney disease, contributing to a number of structural changes that increase the susceptibility of the glomerular basement membrane to barotrauma, contributing to albuminuria, fibrosis, and loss of kidney function. Higher levels of albuminuria might therefore reflect the severity of renal injury from this mechanism. Conversely, others have postulated that changes in vascular (endothelial) function may play a key role.²⁰ In both cases the filtered albumin will increase; it has been proposed that this may lead to direct damage to the glomerulus and the tubule, and ultimately nephron loss.^{21,22} Each of these could be consistent with our observations that the benefits of canagliflozin on renal outcomes, as measured by changes in eGFR slope and the absolute risk reduction for the composite renal outcome, are greatest in people with severely increased albuminuria, the individuals in whom raised glomerular pressure and/or albuminuria are driving the progression of kidney disease.

At the same time, it is recognized that our understanding of the clinical presentation of diabetic kidney disease is evolving.^{23,24} Approximately 40% of people with T2DM have been reported to develop impaired kidney function without ever having albuminuria documented, with a variety of histologic changes on biopsy that appear distinct from individuals with classically progressive albuminuria.25,26 These individuals may have other mechanisms of disease progression. The CANVAS Program data show that canagliflozin appears to stabilize kidney function and prevent the composite kidney outcome even in participants with normal urinary albumin excretion at baseline, despite substantially smaller effects on albuminuria. Mechanisms other than those associated with albuminuria reduction may be important; there is some experimental evidence that SGLT2 inhibitors improve renal oxygenation and promote anti-inflammatory and antifibrotic pathways.²⁷⁻³⁰ Although individuals with T2DM and normal urinary albumin excretion are at low absolute risk for renal outcomes in the short term, the population prevalence is much higher than severely increased albuminuria,^{31,32} and treatment of these individuals with an SGLT2 inhibitor may be an important strategy in reducing the long-term burden of kidney and cardiovascular disease due to T2DM.

Because the absolute risk of events was greater at higher levels of baseline albuminuria, the absolute benefits for renal outcomes and all-cause mortality were greatest in people with severely increased albuminuria. This translated into 5-year numbers needed to treat of seven and ten for these outcomes. Approximately 80% of participants in the CANVAS Program were already receiving RAS blockade at baseline as well as many other preventative therapies, indicating that the renoprotective effect of canagliflozin is achieved in addition to the effects of these agents.

The glycemic efficacy of SGLT2 inhibitors is dependent on kidney function,³³ and differing effects on HbA1c across albuminuria subgroups are likely due to differences in eGFR, which was progressively lower across subgroups with higher baseline albuminuria. The greater proportional reductions in albuminuria in participants with moderately and severely increased albuminuria did not appear to be driven by reductions in systolic BP, which were consistent across subgroups.

Albuminuria has been demonstrated to be among the strongest predictors of cardiovascular outcomes in T2DM.^{34,35} The pathophysiologic link between albuminuria and cardiovascular risk is not entirely understood, but because albuminuria results primarily from injury to glomeruli, it is thought to be a marker of systemic endothelial damage.³⁶ These data suggest that the benefits of canagliflozin for the prevention of cardiovascular outcomes are at least as large in people with elevated levels of albuminuria, and are consistent with a similar analysis of the EMPA-REG OUTCOME trial.¹⁶

Although the risk of amputation and fractures was increased with canagliflozin in the CANVAS Program, no increased risk was observed in the CREDENCE trial.¹⁴ It remains unclear whether this was due to differences in participant characteristics, trial protocols, or chance. Additionally, the risk of fracture was observed in CANVAS but not CANVAS-R, and the reason for heterogeneity in the fracture outcome across albuminuria subgroups also remains unclear.

This study benefits from a number of strengths. These data were derived from a large, multicenter, placebo-controlled, randomized trial program conducted to a high standard. Expert committees blinded to treatment allocations adjudicated all cardiovascular and renal outcomes. The use of continuous eGFR slope data also provided additional power to explore in more detail the renoprotective effect across albuminuria subgroups.

Our findings should be interpreted in light of some limitations. This study was a *post-hoc* analysis and was not explicitly powered to detect cardiovascular or renal benefits in each of the albuminuria subgroups. The reported *P* heterogeneity values were nominal in nature with no corrections applied for multiple comparisons. These *P* values should therefore be interpreted cautiously in light of the number of tests performed. The proportion of participants with severely increased albuminuria was relatively small in comparison to the overall trial population. The high proportion of participants in the CANVAS Program with established cardiovascular disease may limit the generalizability of our findings to the broader population of patients with diabetic kidney disease. Nevertheless, this study represents one of the largest analyses to date of the influence of albuminuria on the effects of SGLT2 inhibition in people with T2DM.

Upcoming trials are expected to provide definitive information about cardiovascular and renal protection for this high-risk population. These include DAPA-CKD for dapagliflozin (NCT03036150),³⁷ EMPA-KIDNEY for empagliflozin (NCT03594110),³⁸ and SCORED with sotagliflozin (NCT03315143).³⁹ Some of these trials will also include participants with eGFR as low as 20 ml/min per 1.73 m², irrespective of levels of albuminuria.

In conclusion, the CANVAS Program data suggest that, among individuals with T2DM, canagliflozin improves renal outcomes. Protective effects on the kidney are observed even in participants with normal albuminuria, with greater absolute benefits among patients with severely increased albuminuria.

ACKNOWLEDGMENTS

We thank all investigators, study teams, and patients for participating in these studies. The authors thank the following individuals for their contributions to the statistical monitoring/analyses and the protocol development, safety monitoring, and operational implementation over the duration of both studies: Lyndal Hones, Sharon Dunkley, Tao Sun, Gordon Law, George Capuano, Severine Bompoint, Laurent Billot, Mary Lee, Joan Lind, Roger Simpson, Mary Kavalam, Ed Connell, Jacqueline Yee, Dainius Balis, Frank Vercruysse, Elisa Fabbrini, Nicole Meyers, Wayne Shaw, Gary Meininger, Mehul Desai, Hsiaowei Deng, and Norm Rosenthal.

The trials were sponsored by Janssen Research & Development, LLC, and were conducted collaboratively by an academic steering committee, an academic research organization, George Clinical, and the sponsor. The sponsor was responsible for study oversight and data collection, and had a representative on the steering committee, which was responsible for study design, data analysis, data interpretation, and writing of this report. All authors had full access to all of the data and had final responsibility for the decision to submit for publication. We report these findings on behalf of the CANVAS Program Collaborative Group.

Medical writing support was provided by Dana Tabor, PhD, of MedErgy, and was funded by Janssen Global Services, LLC. Canagliflozin was developed by Janssen Research & Development, LLC, in collaboration with Mitsubishi Tanabe Pharma Corporation.

Dr. Neuen, Dr. Ohkuma, Mr. Li, and Dr. Oh contributed to the analysis and interpretation of data. Drs. Neal and Matthews were cochairs of the CANVAS Program Steering Committee. Drs. de Zeeuw, Mahaffey, Fulcher, Jardine, and Perkovic contributed to the design and conduct of the study and the interpretation of the data. Drs. Neuen and Perkovic wrote the first draft of the manuscript; all authors contributed to subsequent drafts and approved the final version for submission.

DISCLOSURES

Dr. Neuen is supported by an Australian National Health and Medical Research Council Postgraduate Scholarship; the John Chalmers PhD Scholarship from The George Institute for Global Health; a University Postgraduate Award from University of New South Wales Sydney; the Graduate Research Fund at Lincoln College, University of Oxford; and an Oxford Australia Clarendon Scholarship from the University of Oxford. He has received travel support from Janssen. Dr. Ohkuma is supported by the John Chalmers Clinical Research Fellowship of The George Institute for Global Health. Mr. Li reports being a full-time employee of The George Institute for Global Health. Dr. Neal reports being supported by a National Health and Medical Research Council of Australia Principal Research Fellowship (APP1106947); serving on advisory boards and/or acting as consultant for Janssen and Merck Sharpe and Dohme; and receiving lecture fees from Janssen with any consultancy, honoraria, or travel support paid to his institution. Dr. Matthews reports receiving research support from Janssen; serving on advisory boards and as a consultant for Novo Nordisk, Novartis, Sanofi-Aventis, Janssen, and Servier; and giving lectures for Novo Nordisk, Servier, Sanofi-Aventis, Novartis, Janssen, Mitsubishi Tanabe, and Aché Laboratories. Dr. de Zeeuw reports serving on advisory boards and/ or as a speaker for Bayer, Boehringer Ingelheim, Fresenius, Mundipharma, and Mitsubishi Tanabe; serving on steering committees and/or as a speaker for AbbVie and Janssen; and serving on data safety and monitoring committees for Bayer. Dr. Mahaffey reports grants from Afferent, Amgen, Apple, Inc., AstraZeneca, Cardiva Medical, Inc., Daiichi, Ferring, Google (Verily), Johnson & Johnson, Luitpold, Medtronic, Merck, NIH, Novartis, Sanofi, St. Jude, and Tenax and personal fees from Abbott, Ablynx, AstraZeneca, Baim Institute, Boehringer Ingelheim, Bristol-Myers Squibb, Elsevier, GlaxoSmithKline Johnson & Johnson, MedErgy, Medscape, Mitsubishi, Myokardia, NIH, Novartis, Novo Nordisk, Portola, Radiometer, Regeneron, SmartMedics, Springer Publishing, and UCSF. Dr. Fulcher reports receiving research support from Novo Nordisk and serving on advisory boards and as a consultant for Janssen, Novo Nordisk, Boehringer Ingelheim, and Merck Sharp and Dohme. Dr. Jardine is supported by a Medical Research Future Fund Next Generation Clinical Researchers Program Career Development Fellowship; is responsible for research projects that have received unrestricted funding from Gambro, Baxter, CSL, Amgen, Eli Lilly, and Merck; has served on advisory boards sponsored by Akebia, Baxter, and Boehringer Ingelheim; and has spoken at scientific meetings sponsored by Janssen, Amgen, and Roche, with any consultancy, honoraria, or travel support paid to her institution. Dr. Oh reports being a full-time employee of Janssen Research & Development, LLC. Dr. Perkovic reports receiving research support from the Australian National Health and Medical Research Council (Senior Research Fellowship and Program Grant); serving on steering committees for AbbVie, Boehringer Ingelheim, GlaxoSmithKline, Janssen, and Pfizer; serving on advisory boards and/or speaking at scientific meetings for AbbVie, Astellas, AstraZeneca, Bayer, Baxter, Bristol-Myers Squibb, Boehringer Ingelheim, Durect, Eli Lilly, Gilead, GlaxoSmithKline, Janssen, Merck, Novartis, Novo Nordisk, Pfizer, Pharmalink, Relypsa, Roche, Sanofi, Servier, and Vitae; and receiving personal fees for consulting and scientific presentations related to canagliflozin from Mitsubishi Tanabe and Mundipharma. No other potential conflicts of interest relevant to this article are reported.

FUNDING

This study was supported by Janssen Research & Development, LLC.

SUPPLEMENTAL MATERIAL

This article contains the following supplemental material online at http://jasn.asnjournals.org/lookup/suppl/doi:10.1681/ASN.2019010064/-/DCSupplemental.

Supplemental Table 1. Characteristics of participants with UACR < 30, 30–300, and > 300 mg/g at baseline.

Supplemental Table 2. Effect on total eGFR slope* for the overall trial population and in participants with UACR < 30, 30–300, and > 300 mg/g at baseline.

Supplemental Table 3. Effect of canagliflozin on cardiovascular and renal outcomes by baseline albuminuria adjusted for competing risk of death.

Supplemental Table 4. Trend tests across UACR subgroups (<30, 30-300, and >300 mg/g) and interaction tests using log transformed UACR fitted as a continuous variable for cardiovascular and renal outcomes.

Supplemental Figure 1. Change in intermediate outcomes for canagliflozin and placebo treated participants by UACR (<30, 30-300, and >300 mg/g).

Supplemental Figure 2. Effect of canagliflozin on the (A) primary cardiovascular outcome, (B) hospitalized or fatal heart failure, (C) renal composite outcome, and (D) all-cause mortality in participants stratified into deciles of baseline UACR.

Supplemental Figure 3. Safety outcomes collected in CANVAS alone in participants with UACR < 30, 30–300, and > 300 mg/g at baseline.

REFERENCES

- 1. Alicic RZ, Rooney MT, Tuttle KR: Diabetic kidney disease: Challenges, progress, and possibilities. *Clin J Am Soc Nephrol* 12: 2032–2045, 2017
- 2. Afkarian M, Sachs MC, Kestenbaum B, Hirsch IB, Tuttle KR, Himmelfarb J, et al.: Kidney disease and increased mortality risk in type 2 diabetes. *J Am Soc Nephrol* 24: 302–308, 2013
- van der Velde M, Matsushita K, Coresh J, Astor BC, Woodward M, Levey A, et al.; Chronic Kidney Disease Prognosis Consortium: Lower estimated glomerular filtration rate and higher albuminuria are associated with allcause and cardiovascular mortality. A collaborative meta-analysis of highrisk population cohorts. *Kidney Int* 79: 1341–1352, 2011
- Gansevoort RT, Matsushita K, van der Velde M, Astor BC, Woodward M, Levey AS, et al.; Chronic Kidney Disease Prognosis Consortium: Lower estimated GFR and higher albuminuria are associated with adverse kidney outcomes. A collaborative meta-analysis of general and highrisk population cohorts. *Kidney Int* 80: 93–104, 2011
- Heerspink HJL, Kröpelin TF, Hoekman J, de Zeeuw D; Reducing Albuminuria as Surrogate Endpoint (REASSURE) Consortium: Drug-induced reduction in albuminuria is associated with subsequent renoprotection: A meta-analysis. J Am Soc Nephrol 26: 2055–2064, 2015
- Heerspink HJL, Greene T, Tighiouart H, Gansevoort RT, Coresh J, Simon AL, et al.; Chronic Kidney Disease Epidemiology Collaboration: Change in albuminuria as a surrogate endpoint for progression of kidney disease: A meta-analysis of treatment effects in randomised clinical trials. *Lancet Diabetes Endocrinol* 7: 128–139, 2019
- Heerspink HJ, Perkins BA, Fitchett DH, Husain M, Cherney DZ: Sodium glucose cotransporter 2 inhibitors in the treatment of diabetes mellitus: Cardiovascular and kidney effects, potential mechanisms, and clinical applications. *Circulation* 134: 752–772, 2016
- Heerspink HJL, Kosiborod M, Inzucchi SE, Cherney DZI: Renoprotective effects of sodium-glucose cotransporter-2 inhibitors. *Kidney Int* 94: 26–39, 2018
- Neal B, Perkovic V, Mahaffey KW, de Zeeuw D, Fulcher G, Erondu N, et al.; CANVAS Program Collaborative Group: Canagliflozin and cardiovascular and renal events in type 2 diabetes. N Engl J Med 377: 644–657, 2017
- Perkovic V, de Zeeuw D, Mahaffey KW, Fulcher G, Erondu N, Shaw W, et al.: Canagliflozin and renal outcomes in type 2 diabetes: Results from the CANVAS Program randomised clinical trials. *Lancet Diabetes Endocrinol* 6: 691–704, 2018

- Zinman B, Wanner C, Lachin JM, Fitchett D, Bluhmki E, Hantel S, et al.; EMPA-REG OUTCOME Investigators: Empagliflozin, cardiovascular outcomes, and mortality in type 2 diabetes. N Engl J Med 373: 2117–2128, 2015
- Wanner C, Inzucchi SE, Lachin JM, Fitchett D, von Eynatten M, Mattheus M, et al.; EMPA-REG OUTCOME Investigators: Empagliflozin and progression of kidney disease in type 2 diabetes. N Engl J Med 375: 323–334, 2016
- Wiviott SD, Raz I, Bonaca MP, Mosenzon O, Kato ET, Cahn A, et al.; DECLARE–TIMI 58 Investigators: Dapagliflozin and cardiovascular outcomes in type 2 diabetes. N Engl J Med 380: 347–357, 2019
- Perkovic V, Jardine MJ, Neal B, Bompoint S, Heerspink HJL, Charytan DM, et al.; CREDENCE Trial Investigators: Canagliflozin and renal outcomes in type 2 diabetes and nephropathy. N Engl J Med 380: 2295–2306, 2019
- 15. Fine JP, Gray RJ: A proportional hazards model for the subdistribution of a competing risk. J Am Stat Assoc 94: 496–509, 1999
- Wanner C, Lachin JM, Inzucchi SE, Fitchett D, Mattheus M, George J, et al.; EMPA-REG OUTCOME Investigators: Empagliflozin and clinical outcomes in patients with type 2 diabetes mellitus, established cardiovascular disease, and chronic kidney disease. *Circulation* 137: 119–129, 2018
- 17. Cherney DZI, Zinman B, Inzucchi SE, Koitka-Weber A, Mattheus M, von Eynatten M, et al.: Effects of empagliflozin on the urinary albumin-to-creatinine ratio in patients with type 2 diabetes and established cardiovascular disease: An exploratory analysis from the EMPA-REG OUTCOME randomised, placebo-controlled trial. *Lancet Diabetes Endocrinol* 5: 610–621, 2017
- Wanner C, Heerspink HJL, Zinman B, Inzucchi SE, Koitka-Weber A, Mattheus M, et al.; EMPA-REG OUTCOME Investigators: Empagliflozin and kidney function decline in patients with type 2 diabetes: A slope analysis from the EMPA-REG outcome trial. J Am Soc Nephrol 29: 2755–2769, 2018
- Muskiet MHA, Wheeler DC, Heerspink HJL: New pharmacological strategies for protecting kidney function in type 2 diabetes. *Lancet Diabetes Endocrinol* 7: 397–412, 2019
- Rabelink TJ, de Zeeuw D: The glycocalyx--linking albuminuria with renal and cardiovascular disease. Nat Rev Nephrol 11: 667–676, 2015
- Cravedi P, Remuzzi G: Pathophysiology of proteinuria and its value as an outcome measure in chronic kidney disease. Br J Clin Pharmacol 76: 516–523, 2013
- 22. Roscioni SS, Lambers Heerspink HJ, de Zeeuw D: Microalbuminuria: Target for renoprotective therapy PRO. *Kidney Int* 86: 40–49, 2014
- Thomas MC, Cooper ME, Zimmet P: Changing epidemiology of type 2 diabetes mellitus and associated chronic kidney disease. Nat Rev Nephrol 12: 73–81, 2016
- Afkarian M, Zelnick LR, Hall YN, Heagerty PJ, Tuttle K, Weiss NS, et al.: Clinical manifestations of kidney disease among US adults with diabetes, 1988-2014. JAMA 316: 602–610, 2016
- Retnakaran R, Cull CA, Thorne KI, Adler AI, Holman RR; UKPDS Study Group: Risk factors for renal dysfunction in type 2 diabetes: U.K. Prospective diabetes study 74. *Diabetes* 55: 1832–1839, 2006
- Ekinci El, Jerums G, Skene A, Crammer P, Power D, Cheong KY, et al.: Renal structure in normoalbuminuric and albuminuric patients with type 2 diabetes and impaired renal function. *Diabetes Care* 36: 3620–3626, 2013
- Chang YK, Choi H, Jeong JY, Na KR, Lee KW, Lim BJ, et al.: Dapagliflozin, SGLT2 inhibitor, attenuates renal ischemia-reperfusion injury [published correction appears in *PLoS One* 11: e0160478, 2016]. *PLoS* One 11: e0158810, 2016
- Terami N, Ogawa D, Tachibana H, Hatanaka T, Wada J, Nakatsuka A, et al.: Long-term treatment with the sodium glucose cotransporter 2 inhibitor, dapagliflozin, ameliorates glucose homeostasis and diabetic nephropathy in db/db mice. *PLoS One* 9: e100777, 2014
- Wang XX, Levi J, Luo Y, Myakala K, Herman-Edelstein M, Qiu L, et al.: SGLT2 protein expression is increased in human diabetic nephropathy: SGLT2 protein inhibition decreases renal lipid accumulation,

inflammation, and the development of nephropathy in diabetic mice. *J Biol Chem* 292: 5335–5348, 2017

- 30. O'Neill J, Fasching A, Pihl L, Patinha D, Franzén S, Palm F: Acute SGLT inhibition normalizes O2 tension in the renal cortex but causes hypoxia in the renal medulla in anaesthetized control and diabetic rats. Am J Physiol Renal Physiol 309: F227–F234, 2015
- Pugliese G, Solini A, Bonora E, Fondelli C, Orsi E, Nicolucci A, et al.; RIACE Study Group: Chronic kidney disease in type 2 diabetes: Lessons from the Renal Insufficiency And Cardiovascular Events (RIACE) Italian multicentre study. Nutr Metab Cardiovasc Dis 24: 815–822, 2014
- 32. Yokoyama H, Sone H, Oishi M, Kawai K, Fukumoto Y, Kobayashi M; Japan Diabetes Clinical Data Management Study Group: Prevalence of albuminuria and renal insufficiency and associated clinical factors in type 2 diabetes: the Japan Diabetes Clinical Data Management study (JDDM15). Nephrol Dial Transplant 24: 1212–1219, 2009
- Neuen BL, Ohkuma T, Neal B, Matthews DR, de Zeeuw D, Mahaffey KW, et al.: Cardiovascular and renal outcomes with canagliflozin according to baseline kidney function. *Circulation* 138: 1537–1550, 2018

- Ninomiya T, Perkovic V, de Galan BE, Zoungas S, Pillai A, Jardine M, et al.; ADVANCE Collaborative Group: Albuminuria and kidney function independently predict cardiovascular and renal outcomes in diabetes. J Am Soc Nephrol 20: 1813–1821, 2009
- 35. Matsushita K, Coresh J, Sang Y, Chalmers J, Fox C, Guallar E, et al.; CKD Prognosis Consortium: Estimated glomerular filtration rate and albuminuria for prediction of cardiovascular outcomes: A collaborative meta-analysis of individual participant data. *Lancet Diabetes Endocrinol* 3: 514–525, 2015
- Weir MR: Microalbuminuria and cardiovascular disease. Clin J Am Soc Nephrol 2: 581–590, 2007
- Pecoits-Filho R, Perkovic V: Are SGLT2 inhibitors ready for prime time for CKD? Clin J Am Soc Nephrol 13: 318–320, 2018
- Herrington WG, Preiss D, Haynes R, von Eynatten M, Staplin N, Hauske SJ, et al.: The potential for improving cardio-renal outcomes by sodiumglucose co-transporter-2 inhibition in people with chronic kidney disease: A rationale for the EMPA-KIDNEY study. *Clin Kidney J* 11: 749–761, 2018
- 39. Sanofi. Effect of sotagliflozin on cardiovascular and renal events in patients with type 2 diabetes and moderate renal impairment who are at cardiovascular risk, 2017. Available at: https://clinicaltrials.gov/ct2/show/ NCT03315143. Accessed September 2019